Author's response to reviews

Title: Perceptions, attitudes and knowledge of evidence-based medicine in primary care: a study protocol

Authors:

- Pablo Alonso-Coello (palonso@santpau.cat)
- Ivan Solà (Isola@santpau.cat)
- Rafael Rotaech (rrotaechec@terra.es)
- Ana I González-González (anaglezglez@gmail.com)
- Mercè Marzo-Castillejo (mmarzoc@gencat.net)
- Arturo Louro-González (alourog@gmail.com)
- Ricard Carrillo (25681rcm@comb.es)
- Paola Velazquez (paolavp@gmail.com)
- Guillermo García-Velasco (garciaav@medynet.com)
- Carlos Calderon (ccalderong@telefonica.net)

Version: 3 Date: 12 May 2009

Author's response to reviews: see over
Dear Sir/Madam,

We have addressed the following comments made to our manuscript:

- We have updated the author list and added a corresponding author. We have included all e-mails in the correct format.
- We have updated the references in the format suggested.
- We have corrected reference 15-0 and the rest of the changes suggested in the revision related to them.
- In reference 14 we have updated the text and added a new link.
- We have changed the link to reference 29
- We have renamed the design section.
- We have included the abbreviations as a paragraph and included sources of funding in the Acknowledgements section. We have moved references to after Acknowledgements.

Yours sincerely,

Pablo Alonso-Coello